Author’s response to reviews

Title: In vitro Anti-mycobacterial activity of Selected Medicinal Plants against Mycobacterium tuberculosis and Mycobacterium bovis Strains

Authors:

Abdella Gemechu (abdgemechu@gmail.com)
Mirutse Giday (mirutseg@gmail.com)
Gobena Ameni (gobenachimdi2009@yahoo.co.uk)
Adane Worku (adanewrk@yahoo.com)

Version: 2 Date: 29 August 2013

Author's response to reviews: see over
Title: In vitro Anti-mycobacterial activity of Selected Medicinal Plants against Mycobacterium tuberculosis and Mycobacterium bovis Strains

Version: 1 Date: 21 Aug 2013

Dear Editor-in-Chief:

We wish to submit a manuscript entitled “In vitro Anti-mycobacterial activity of Selected Medicinal Plants against Mycobacterium tuberculosis and Mycobacterium bovis Strains” as a manuscript explaining suitability of our study for future treatment interventions. On the other hand this study will serve as source of information on the efficacy these medicinal plants in treating TB and/ or respiratory illness, while contributing to research development. The significance of this report is that it contributes to the understanding of suitable areas where more studies are needed to understand efficacy of these medicinal plants in treatment particularly in TB and other respiratory disorders. Moreover, it can be used as baseline information for further phyto-chemical, toxicity and pharmaceutical studies.

The material in this manuscript is original and has not been published. Moreover this material has not and will not be submitted elsewhere for publication as long as it is still under consideration by the journal, and if accepted by your journal will not be published elsewhere. This manuscript has two tables. None of the authors have conflict of interest.

We Abdella Gemechu, Mirutse Giday, Gobena Ameni and Adane Worku - believe that your journal is the most appropriate for the publication of this manuscript and we hope that your reviewers will find it suitable. Thus, we will look forward for positive response.

Yours Sincerely,

Abdella Gemechu

Corresponding Author
Haramaya University, Health and Medical Science College.